---
title: "Introduction"
description: "Welcome to the Trigger.dev v3 documentation."
---

<Warning>
  The Trigger.dev v3 Developer Preview is currently in invite-only early access. [Sign up here to
  request access](https://trigger.dev/v3-early-access).
</Warning>

## What is Trigger.dev (v3)?

Trigger.dev v3 makes it easy to write reliable long-running tasks without timeouts.

- We run your tasks with no timeouts. You don't have to manage any infrastructure (unless you [self-host](/v3/open-source-self-hosting)). Workers are automatically scaled and managed for you.
- We provide a multi-tenant queue that is used when triggering tasks.
- We provide an SDK and CLI for writing tasks in your existing codebase, inside [/trigger folders](/v3/trigger-folder).
- We provide different types of tasks: [regular](/v3/tasks-regular), [scheduled](/v3/tasks-scheduled), [zod](/v3/tasks-zod), [webhooks](/v3/tasks-webhooks).
- We provide a dashboard for monitoring, debugging, and managing your tasks.

We're [open source](https://github.com/triggerdotdev/trigger.dev) and you can choose to use the [Trigger.dev Cloud](https://cloud.trigger.dev) or [Self-host Trigger.dev](/v3/open-source-self-hosting) on your own infrastructure.

## Getting started

<CardGroup>
  <Card title="Quick start guide" icon="person-running-fast" href="/v3/quick-start">
    Get started in 3 minutes.
  </Card>
  <Card title="Writing tasks" icon="wand-magic-sparkles" href="/v3/tasks-overview">
    Tasks are the core of Trigger.dev. Learn what they are and how to write them.
  </Card>
</CardGroup>

## Getting help

We'd love to hear from you or give you a hand getting started. Here are some ways to get in touch with us. We'd also ❤️ your support.

<CardGroup>
  <Card
    title="Join our Discord server"
    icon="discord"
    href="https://discord.gg/kA47vcd8P6"
    color="#5865F2"
  >
    The help forum is a great place to get help with any questions about Trigger.dev.
  </Card>
  <Card
    title="Follow us on X (Twitter)"
    icon={
      <svg xmlns="http://www.w3.org/2000/svg" height="20" viewBox="0 0 512 512">
        <path d="M389.2 48h70.6L305.6 224.2 487 464H345L233.7 318.6 106.5 464H35.8L200.7 275.5 26.8 48H172.4L272.9 180.9 389.2 48zM364.4 421.8h39.1L151.1 88h-42L364.4 421.8z" />
      </svg>
    }
    href="https://twitter.com/triggerdotdev"
    color="#1DA1F2"
  >
    Follow us on X (Twitter) to get the latest updates and news.
  </Card>
  <Card
    title="Schedule a call"
    icon="phone"
    iconType="solid"
    href="https://cal.com/team/triggerdotdev/founders-call"
  >
    Arrange a call with one of the founders. We can help answer questions and give 1-on-1 help
    building your first task.
  </Card>
  <Card
    title="Give us a star on GitHub"
    icon="star"
    iconType="solid"
    href="https://github.com/triggerdotdev/trigger.dev"
    color="#fbbf24"
  >
    Check us out at triggerdotdev/trigger.dev
  </Card>
</CardGroup>
